Author’s response to reviews

Title: Relationship between inflammatory markers and mild cognitive impairment in Chinese patients with type 2 diabetes: a case-control study

Authors:
Miaoyan Zheng (myzheng73@163.com)
Baocheng Chang (changbc1970@126.com)
Liqiang Tian (defjmmm@163.com)
Chunyan Shan (chunyanshan@hotmail.com)
Hui Chen (602090529@qq.com)
Yuxia Gao (gaoyuxia@medmail.com.cn)
Guowei Huang (tjgwhuang@126.com)
Meilin Zhang (zhangmeilin@tmu.edu.cn)

Version: 3 Date: 13 Jun 2019

Author’s response to reviews:

Dear Professor Ciarán Martin Fitzpatrick,

Thank you for the opportunity to improve our manuscript (BEND-D-19-00016) " . We have revised the manuscript carefully according to the comments of the reviewers. We provide below our detailed responses to your comments.

Thank you for your editorial assistance. We hope that the revised manuscript is found acceptable for publication in your journal. If you have any questions about our revised manuscript, please inform us without hesitation.

Sincerely Yours,

Meilin Zhang
Department of Nutrition and Food Science
School of Public Health
The manuscript has been improved substantially after the revisions. However, there are a lot of mistakes in the characters in the revised version.
Response: Thank for your consideration. We have modified the mistakes in the characters carefully in the revised version.

Reviewer reports:
So Nagai, M.D., Ph. D. (Reviewer 1): None

Hiroshi Akasaka (Reviewer 2): Authors revised the manuscript each comment that I have pointed out, and the revised version is satisfactory. I think that the revised version is now suitable for publication except only one point: in Figure 1, two "from" overlap.
Response: Thank for your consideration. We have deleted the overlap word in Figure 1.